Author’s response to reviews

Title: Re-growth of an incomplete discoid lateral meniscus after arthroscopic partial resection in an 11 year-old boy: a case report

Authors:

Salvatore Bisicchia (s.bisicchia@gmail.com)
Cosimo Tudisco (cosimo.tudisco@uniroma2.it)

Version: 2 Date: 28 August 2013

Author’s response to reviews: see over
Dear Editor,

Please accept the revised manuscript. The manuscript has been revised by a native English speaker.

In reference to the reviewers comments:

Line 37: we agree. We added the following sentence to report the frequency of discoid meniscus: “The prevalence of discoid lateral meniscus ranges from 0.4 to 17%”.

Line 54: we agree. We corrected the sentence as follows: “leaving a functional residual rim of 8 mm”.

Line 60: we agree. We corrected the sentence as follows: “In the meantime, he was going through a growth spurt and he had grown about 15 cm since the first surgery”.